Author’s response to reviews

Title: Homocysteine, Methylenetetrahydrofolate Reductase C677T Polymorphism, and Risk of Retinal Vein Occlusion: An Updated Meta-analysis

Authors:

Minwen Zhou (zmw8008@163.com)
Xiaodong Sun (466741043@qq.com)
Xun Xu (563591810@qq.com)
Dan li (zmw8006@163.com)

Version: 4 Date: 20 October 2014

Author’s response to reviews:

Dear Dr. Cruz,

Enclosed please find the revised version of our manuscript “Homocysteine, Methylenetetrahydrofolate Reductase C677T Polymorphism, and Risk of Retinal Vein Occlusion: An Updated Meta-analysis” (Manuscript ID: 2138316479137220).

We appreciate your supportive comments and the Reviewers’ thoughtful and constructive critiques, which had been extremely helpful in our revision. We have considered each of the reviewer’s comments, and have responded carefully and thoroughly as summarized in the following pages, which we believe that the article is thereby strengthened.

Thank you again for your further consideration of this manuscript.

Sincerely yours,
Minwen Zhou, MD

"There is one point need further revision. The term "trials" is usually used to describe treatment study. Since the original studies included are observational studies, please change the term "trial" to "study".

Answer: We thank-you for this insightful suggestion. As advised, we have changed the term "trial" to "study". Thank you!"